Author's response to reviews

Title: Modular prevention of heart disease following acute coronary syndrome (ACS) [ISRCTN42984084]

Authors:

Julie Redfern (julieredfern@dodo.com.au)
Elizabeth R Ellis (elizabethellis@ozemail.com.au)
Tom Briffa (T.Briffa@curtin.edu.au)
Saul B Freedman (ben@gmp.usyd.edu.au)

Version: 3 Date: 2 June 2006

Author's response to reviews:

Guided self-choice management of heart disease study

2st June 2006

Dear Editor BMC Cardiovascular Disorders,

Please find attached our revised study protocol entitled "Modular prevention of heart disease following acute coronary syndrome (ACS) [ISRCTN42984084]". Point-by-point changes made are listed below.

1. There are no additional files referred to in the text and hence have been removed from the submissions page.

Thank you for your feedback.

Sincerely,

Julie Redfern (on behalf of authors)